Author’s response to reviews

Title: Self-Reported Hypoglycemia in Adult Diabetic Patients in East Gojjam, Northwest Ethiopia: Institution Based Cross-Sectional Study

Authors:

Gashayeneh Genetu Tiruneh (gashayeneh1@gmail.com)
Nurilign Abebe (nure113@gmail.com)
Getenet Dessie (ayalew.d16@gmail.com)

Version: 1 Date: 19 Jun 2018

Author’s response to reviews:

Response to editors:

1. Re-phrase the background section
   Re: Done

2. Provide figure legends under separate heading
   Re: Done

3. Amend heading of 'Conflict of interests' to 'Competing interests'
   Re: Done

4. Upload manuscript as final version
   Re: Done